Author's response to reviews

Title: Designing a theory-informed, contextually appropriate intervention strategy to improve delivery of paediatric services in Kenyan hospitals.

Authors:

Mike English (menglish@nairobi.kemri-wellcome.org)

Version: 3 Date: 15 March 2013

Author's response to reviews:

Hi, I have reviewed the changes made at copy editing stage and accepted them and addressed the few comments from the copy editors in the clean version submitted.

In the manuscript, p 17 para 1, I refer to a paper in preparation by Nzinga et al. This paper is now published; Nzinga J, Mbaabu L, English M: Service delivery in Kenyan district hospitals -- what can we learn from literature on mid-level managers? Human resources for health 2013, 11(1):10.

I have been unable to update the bibliography as for some reason the numbering of other references then seems to become confused! Not sure why this is happening and perhaps easier to be done at the copyediting? If not I can keep trying,

Mike